Author's response to reviews

Title: Physical activity, screen time and obesity status in a nationally representative sample of Maltese youth with international comparisons

Authors:

Andrew Decelis (andrew.decelis@um.edu.mt)
Russell Jago (russ.jago@bristol.ac.uk)
Kenneth R. Fox (K.R.Fox@bristol.ac.uk)

Version: 7 Date: 6 June 2014

Author's response to reviews: see over
5\textsuperscript{th} June, 2014

Dear Dr. Clemes,

We would like to thank you for the reviews of our paper \textit{Physical activity, screen viewing and obesity status in a nationally representative sample of Maltese youth with international comparisons}. In light of the comments we have amended the paper and highlighted the changed sections in yellow. Please find a point by point response to the reviews below.

Sincerely

Andrew Decelis
o.b.o co-authors Prof. Russell Jago & Prof. Kenneth Fox

(Reviewer: Dr. Rezende, 3 June 2014)

\textit{References}

- The references 15 and 45 seems to be the same.

Extra reference has been removed.

\textit{Introduction}

-\textit{I still suggest the authors to review the 2nd paragraph that remains too long (15 lines long) and presents different concepts and ideas together.}

We have divided the paragraph into three shorter paragraphs separating different concepts and ideas, while editing some parts of these paragraphs (Lines 64-82).